\documentclass[12pt, leqno]{article}
\usepackage{indentfirst}
%\documentstyle[12pt]{amsart}
%\usepackage{amssymb,amsmath}\textbf{}
\usepackage{amstext}
\usepackage{amsthm}
\usepackage{amsopn}
\usepackage{amsfonts}
\usepackage{amsmath}
\usepackage{latexsym}
\usepackage{amscd}
\usepackage{amssymb}
\usepackage{amsmath}

%\newcommand{\fref}[1]{Figure~\ref{fig:#1}}

%\textheight     =8.85in
\textheight     =8in
\topmargin      =-.1in          % LaTeX uses too much space on top


\textwidth      =6in \oddsidemargin  =.25in \evensidemargin
=\oddsidemargin \font\teneufm=eufm10 \font\seveneufm=eufm7
\font\fiveeufm=eufm5
\newfam\eufmfam 
\textfont\eufmfam=\teneufm \scriptfont\eufmfam=\seveneufm
\scriptscriptfont\eufmfam=\fiveeufm
\def\frak#1{{\fam\eufmfam\relax#1}}
\let\goth\mathfrak
\let\bb\mathbb
\def\gg{\goth g}
\def\gh{\goth h}
\def\gk{\goth k}
\def\gl{\goth l}
\def\gi{\goth i}
\def\gc{\goth c}
\def\gb{\goth b}
\def\gp{\goth p}
\def\gr{\goth r}
\def\gg{\goth g}
\def\gu{\goth u}
\def\gq{\goth q}
\def\gz{\goth z}
\def\go{\goth o}
\def\ga{\goth a}
\def\gd{\goth d}
\def\sl{\goth{s}\goth{l}}
\def\pp{\mbox{\bf p}}
\def\1{\mbox{\bf 1}}
\def\psp{\mbox{\bf sp}}

% LaTeX begin and end of numbered equation
%

\newcommand{\V}{{\mathfrak V}}
\newcommand{\E}{{\mathfrak E}}


\DeclareMathOperator{\supp}{supp} \DeclareMathOperator{\Span}{Span}
\DeclareMathOperator{\tr}{tr} \DeclareMathOperator{\Hom}{Hom}
\DeclareMathOperator{\Aut}{Aut} \DeclareMathOperator{\Int}{Int}
\DeclareMathOperator{\Out}{Out} \DeclareMathOperator{\Der}{Der}
\DeclareMathOperator{\im}{im} \DeclareMathOperator{\ch}{ch}
\DeclareMathOperator{\OR}{\bf O}
\DeclareMathOperator{\cd}{\rm cd}

\DeclareMathOperator{\br}{\rm Br}
\DeclareMathOperator{\sd}{\rm sd}
%\DeclareMathOperator{\char}{char}
\DeclareMathOperator{\End}{End} \DeclareMathOperator{\Id}{Id}
%\DeclareMathOperator{\deg}{deg}
%\DeclareMathOperator{\det}{det}
\DeclareMathOperator{\Sup}{Sup} \DeclareMathOperator{\Ind}{Ind}
\DeclareMathOperator{\Ad}{Ad} 
\DeclareMathOperator{\Spin}{\bf Spin}
\DeclareMathOperator{\Sp}{\bf Sp}
\DeclareMathOperator{\SO}{\bf SO} 
\DeclareMathOperator{\PGL}{\bf PGL} 
\DeclareMathOperator{\GL}{\bf GL} 
\DeclareMathOperator{\SL}{\bf
SL} \DeclareMathOperator{\Cliff}{\bf Cliff}
\DeclareMathOperator{\Gr}{Gr}
\newcommand{\ihom}{{\underline{\rm Hom }}}

%*******
\newtheorem{theorem}{Theorem}[section]%Theorems et al in italic font.
\newtheorem{acknowledgement}[theorem]{Acknowledgement} 
\newtheorem{algorithm}[theorem]{Algorithm}
\newtheorem{axiom}[theorem]{Axiom}
\newtheorem{case}[theorem]{Case}
\newtheorem{claim}[theorem]{Claim}
\newtheorem{conclusion}[theorem]{Conclusion}
\newtheorem{condition}[theorem]{Condition}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{criterion}[theorem]{Criterion}
%\newtheorem{definition}[theorem]{Definition}
\newtheorem{exercise}[theorem]{Exercise}
\newtheorem{lemma}[theorem]{Lemma}
%\newtheorem{notation}[theorem]{Notation}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{fact}[theorem]{Fact}
\newtheorem{example}[theorem]{Example}
\newtheorem{examples}[theorem]{Examples}


%\theorembodyfont{\upshape}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{remarks}[theorem]{Remarks}

\newtheorem{question}[theorem]{Question}
\newtheorem{solution}[theorem]{Solution}
\newtheorem{summary}[theorem]{Summary}
%\newenvironment{proof}[1][Proof]{\noindent\textbf{#1.} }{\ \rule{0.5em}{0.5em}}
\newtheorem*{assumption}{Assumption} %Added BNA

\theoremstyle{definition}%Theorems et al in roman font.

\newtheorem{notation}[theorem]{Notation}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{construction}[theorem]{Construction}

\numberwithin{equation}{section}

%*******
%\def\N{\Bbb N}
\def\Z{\mathbb Z}
\def\C{\mathbb C}
\def\R{\mathbb R}
\def\Q{\mathbb Q}
%\def\P{\Bbb P}
%\def\sv{{\sl v}}
\def\N{\mathbb N}
\def\Z{\mathbb Z}
\def\G{\mathbb G}
\def\A{\mathbb A}
\def\C{\mathbb C}
\def\R{\mathbb R}
%\def\Q{\mathbb Q}insi,
\def\bB{\text{\rm \bf B}}
\def\bC{\text{\rm \bf C}}
\def\bE{\text{\rm \bf E}}
\def\bF{\text{\rm \bf F}}
\def\bH{\text{\rm \bf H}}
\def\bG{\text{\rm \bf G}}
\def\bN{\text{\rm \bf N}}
\def\bU{\text{\rm \bf U}}
\def\bV{\text{\rm \bf V}}

\def\bW{\text{\rm \bf W}}
\def\bZ{\mathcal{Z}}

\def\pr{\prime}
\def\lra{\longrightarrow}

\def\bX{\text{\rm \bf X}}
\def\CL{{\cal L}}
\def\CB{{\cal B}}
\def\CM{{\cal M}}
\def\CR{{\cal R}}
\def\Ctd{\text{\rm Ctd}}
\def\Mult{\text{\rm Mult}}
\def\ad{\text{\rm ad}}
\def\bGL{\text{\rm \bf GL}}
\def\bPGL{\text{\rm \bf PGL}}
\def\bSL{\text{\rm \bf SL}}
\def\bT{\text{\rm \bf T}}
\def\bS{\text{\rm \bf S}}
\def\bL{\text{\rm \bf L}}
\def\L{\mathcal{L}}
\def\bQ{\text{\rm \bf Q}}
\def\bP{\text{\rm \bf P}}
\def\bM{\text{\rm \bf M}}
\def\bY{\text{\rm \bf Y}}
\def\bK{\text{\rm \bf K}}
\def\P{\mathbb P}
\def\ba{\pmb a}
\def\be{\mathbf e}
\def\bg{{\bf g}}
\def\bm{\mathbf m}
\def\bn{\mathbf n}
\def\bt{\mathbf t}
\def\bs{{\pmb\sigma}}
\def\bx{{\pmb x}}
\def\by{{\pmb y}}
\def\bt{{\pmb\tau }}
\def\bmu{{\pmb\mu }}
\def\wh{\widehat}
\def\wt{\widetilde}
\def\us{\underset}
\def\os{\overset}
\def\ol{\overline}
\def\id{\text{\rm id}}
\def\alg{\text{\rm alg}}
\def\mod{\text{\rm mod}}
\def\q{\quad}
\def\vs{\vskip.3cm}
\def\vsk{\vskip.5cm}
\def\noi{\noindent}
\def\Lie{\mathop{\rm Lie}\nolimits}
\def\idotsint{\mathop{\int\cdots\int}\nolimits}
\def\sumsum{\mathop{\sum\sum}\nolimits}
\def\limsup{\mathop{\lim\sup}\nolimits}
\def\2int{\mathop{2\int}\nolimits}
\def\Dyn{\text{\bf Dyn}}
\def\Div{\mathop{\rm Div}\nolimits}
\def\End{\mathop{\rm  End}\nolimits}
\def\rank{\mathop{\rm rank}\nolimits}
\def\dim{\mathop{\rm dim}\nolimits}
\def\codim{\mathop{\rm codim}\nolimits}
\def\Spec{\mathop{\rm Spec}\nolimits}
\def\Sup{\mathop{\rm Sup}\nolimits}
\def\Lie{\mathop{\rm Lie}\nolimits}
\def\Ext{\mathop{\rm Ext}\nolimits}
\def\Hom{\mathop{\rm Hom}\nolimits}
\def\Der{\mathop{\rm Der}\nolimits}
\def\Stab{\mathop{\rm Stab}\nolimits}
\def\Inf{\mathop{\rm Inf}\nolimits}
\def\Ind{\mathop{\rm Ind}\nolimits}
\def\mod{\mathop{\rm mod}\nolimits}
\def\link{\mathop{\rm link}\nolimits}
\def\Mod{\mathop{\rm Mod}\nolimits}
\def\Gal{\mathop{\rm Gal}\nolimits}
\def\Irr{\mathop{\rm Irr}\nolimits}
\def\Int{\mathop{\rm Int}\nolimits}
\def\nrd{\mathop{\rm nrd}\nolimits}
\def\Nrd{\mathop{\rm nrd}\nolimits}
\def\Div{\mathop{\rm Div}\nolimits}
\def\car{\mathop{\rm car}\nolimits}
\def\Pic{\mathop{\rm Pic}\nolimits}
\def\Br{\mathop{\rm Br}\nolimits}
\def\Aut{\text{\bf{Aut}}}
\def\Out{\text{\bf{Out}}}
\def\Int{\mathop{\rm Int}\nolimits}
\def\Rep{\mathop{\rm Rep}\nolimits}
\def\CRIS{\mathop{\rm CRIS}\nolimits}
\def\Isom{\mathop{\rm Isom}\nolimits}
\def\Frob{\mathop{\rm Frob}\nolimits}
\def\resp.{\mathop{\rm resp.}\nolimits}
\def\limproj{\mathop{\oalign{lim\cr
\hidewidth$\longleftarrow$\hidewidth\cr}}}
\def\limind{\mathop{\oalign{lim\cr
\hidewidth$\longrightarrow$\hidewidth\cr}}}
\def\Ker{\mathop{\rm Ker}\nolimits}
\def\proj{\mathop{\rm proj}\nolimits}
\def\Im{\mathop{\rm Im}\nolimits}
\def\Cok{\mathop{\rm Cok}\nolimits}
\def\Res{\mathop{\rm Res}\nolimits}
\def\Cor{\mathop{\rm Cor}\nolimits}
\def\bfmu{\hbox{\bfmath\char'26}}
\def\lgr{\longrightarrow}
\def\la{\longleftarrow}
\def\rg{\rightarrow}
\def\lmp{\longmapsto}
\def\card{\mathop{\rm card}\nolimits}
\font\math=cmmi10
\def\varpi{\hbox{\math\char'44}}
\def\simlgr{\buildrel\sim\over\lgr}
\def\simla{\buildrel\sim\over\la}
\def\Coker{\mathop{\rm Coker}\nolimits}
\def\Sym{\mathop{\rm Sym}\nolimits}
\def\Jac{\mathop{\rm Jac}\nolimits}
\def\pa{\S\kern.15em }
\def\tvi{\vrule height 10pt depth 5pt width 0pt}
\def\tv{\tvi\vrule}
\def\clo{\mathrel\bullet\joinrel\mathrel-\joinrel
\mathrel-\joinrel\mathrel\bullet}
\def\cloz{\mathrel\bullet\joinrel\mathrel-\joinrel
\mathrel-\joinrel\mathrel\bullet\joinrel\mathrel-
\joinrel\mathrel-\joinrel\mathrel\bullet}
\def\clos{\mathrel\bullet\joinrel\mathrel-\joinrel
\mathrel-\joinrel\mathrel\bullet\joinrel\mathrel-insi,
\joinrel\mathrel-\joinrel\mathrel\bullet\joinrel\mathrel-
\joinrel\mathrel-\joinrel\mathrel\bullet}
\def\E{{\mathfrak E}}
\def\GB{{\mathfrak G}}
\def\PB{{\mathfrak P}}

\def\tr{\mathop{\rm tr}\nolimits}
%\def\rtimes{\mathbin{{>}\!{\triangleleft}}}
\def\un{\uppercase\expandafter{\romannumeral 1}}
\def\deux{\uppercase\expandafter{\romannumeral 2}}
\def\trois{\uppercase\expandafter{\romannumeral 3}}
\def\quatre{\uppercase\expandafter{\romannumeral 4}}
\def\cinq{\uppercase\expandafter{\romannumeral 5}}
\def\six{\uppercase\expandafter{\romannumeral 6}}
\def\eskip{ \hskip .2 em}
\def\cskip{ \hskip -.2 em}
\def\gg{\goth g}
\def\et{\acute et}


\def\hfl#1#2#3{\smash{\mathop{\hbox to#3{\rightarrowfill}}\limits
^{\scriptstyle#1}_{\scriptstyle#2}}}
\def\gfl#1#2#3{\smash{\mathop{\hbox to#3{\leftarrowfill}}\limits
^{\scriptstyle#1}_{\scriptstyle#2}}}
\def\vfl#1#2#3{\llap{$\scriptstyle #1$}
\left\downarrow\vbox to#3{}\right.\rlap{$\scriptstyle #2$}}
\def\ufl#1#2#3{\llap{$\scriptstyle #1$}
\left\uparrow\vbox to#3{}\right.\rlap{$\scriptstyle #2$}}
\def\pafl#1#2#3{\llap{$\scriptstyle #1$}
\left\Vert\vbox to#3{}\right.\rlap{$\scriptstyle #2$}}
\def\vl#1#2#3{\llap{$\scriptstyle #1$}
\left\vert\vbox to#3{}\right.\rlap{$\scriptstyle #2$}}

\def\kalg{k\text{--}alg}

\def\kfibre{\mathop{\oalign{$\times$\cr
\hidewidth${^k}$\hidewidth\cr}}}


\begin{document}
\begin{thebibliography}{99}

\bibitem{Ar} J. Kr. Arason, {\it  Cohomologische invarianten quadratischer Formen},  J. Algebra  {\bf 36}  (1975),  448--491. 

\bibitem{A} M. Artin, {\it
Brauer-Severi varieties},   Brauer groups in ring theory and algebraic geometry (Wilrijk, 1981),
Lecture Notes in Math. {\bf  917} (1982), 
 194--210,  Springer.


%\bibitem{BS} P. Barquero-Salavert, 
%{\it Similitudes of algebras with involution under odd-degree extensions},   Comm. Algebra {\bf  34}  (2006), 625--632. 

\bibitem{Ba} E. ~Bayer-Fluckiger,   {\it 
Galois cohomology of the classical groups},  Quadratic forms and their applications (Dublin, 1999), 1--7,
Contemp. Math. {\bf  272} (2000), Amer. Math. Soc. 

\bibitem{BL} E. ~Bayer-Fluckiger and H.W. Lenstra Jr., {\it Forms in odd degree extensions and
self-dual normal bases}, Amer. J. Math. {\bf 112} (1990), 359--373.

\bibitem{BP} E. ~Bayer-Fluckiger and    R. ~Parimala,   {\it  Galois cohomology of the classical groups over fields of
cohomological dimension $\leq 2$},  Invent. Math. {\bf 122} (1995), 195--229.


%\bibitem{BP2} E. ~Bayer-Fluckiger  and    R. ~Parimala,   {\it  Classical groups and the Hasse principle},
 % Annals math. {\bf 147} (1998), 651-693.


\bibitem{BFT} G. Berhuy, C. Frings and J.-P. Tignol, {\it Serre's conjecture II
for classical groups over imperfect fields}, J. of Pure and Applied Algebra {\bf 211} (2007), 307-341.



\bibitem{Bo} A. Borel, {\it  Linear Algebraic Groups (Second enlarged edition)},
Graduate text in Mathematics {\bf 126} (1991), Springer.


\bibitem{BoS}  A.  Borel and  T.A. Springer,
  {\it Rationality properties of linear algebraic group II}, T\^ohoku Math. J. {\bf 20}, (1968), 443--497.



\bibitem{BoT}  A.  Borel and  J.  Tits,
  {\it Groupes r\'eductifs}, Pub. Math. IHES {\bf 27}, (1965), 55--152.


 %\bibitem{Br}  N. Bourbaki, {\it  Groupes et alg\`ebres de Lie, Ch. 4, 5 et 6}, 
%Hermann, Paris, (1968).

\bibitem{BT}  A. Borel  and J. Tits,   {\it Groupes r\'eductifs},
    Inst. Hautes \'Etudes Sci. Publ. Math. {\bf  27} (1965),
 55--150.


\bibitem{BKG} M. Borovoi,  B. Kunyavski\v {\i}, and P.  Gille, {\it  Arithmetical birational 
invariants of linear algebraic groups over two-dimensional geometric fields},   J. Algebra  {\bf 276}  (2004),  292--339. 

\bibitem{BT1} F. Bruhat  and  J. Tits, {\it Groupes r\'eductifs sur un corps local. I},
 Inst. Hautes Etudes Sci. Publ. Math.  {\bf  41}  (1972), 5--251.

\bibitem{BT2} F. Bruhat and J. Tits,  {\it Groupes r\'eductifs sur un corps local.
 II. Sch\'emas en groupes. Existence d'une donn\'ee radicielle valu\'ee},
 Inst. Hautes Etudes Sci. Publ. Math.  {\bf  60}  (1984), 197--376.

\bibitem{BT3} F. Bruhat and J. Tits,  {\it Groupes r\'eductifs sur un corps local.
 Chap. III. Compl\'ement et applications \`a la cohomologie galoisienne},
 J. Fasc. Sci. Tokyo {\bf 34} (1987), 671--698.


\bibitem{C89} V. ~Chernousov,  {\it  The Hasse principle for
groups of type $E_8$}, Dokl. Akad. Nauk. SSSR {\bf 306} (1989),
1059-1063 and english transl. in  Math. USSR-Izv. {\bf 34} (1990), 409-423.

\bibitem{C94} V. ~Chernousov,  {\it  Remark on the Serre mod-$5$ invariant
for groups of type $E_8$}, Math. Zametki {\bf 56} (1994), 116--121,
Eng. translation Math. Notes {\bf 56} (1994), 730--733 



\bibitem{C03} V. ~Chernousov,  {\it  The kernel of the Rost invariant, Serre's conjecture II and 
the Hasse principle for quasi-split groups ${}\sp {3,6}D\sb 4,E\sb 6,E\sb 7$},
Math. Ann. {\bf 326} (2003) , 297--330. 

\bibitem{CGR} V. ~Chernousov, P. Gille and Z. Reichstein, {\it 
 Resolution of torsors by abelian  extensions},  Journal of Algebra 296 (2006), 561-581.

\bibitem{CP} V. ~Chernousov and V. P. ~Platonov, {\it  The rationality problem for semisimple group varieties}, 
  J. reine angew. math. {\bf  504}  (1998), 1--28.


%\bibitem{CT96} J.-L. ~Colliot-Th\'el\`ene,  {\it  Groupes lin\'eaires sur les corps de fonctions de courbes r\'eelles},
% J. f\"ur die reine und angew. Mathematik {\bf 474} (1996), 139-167.

\bibitem{CTdJ} J.-L. Colliot-Th\'el\`ene,  {\it 
Alg\`ebres simples centrales sur les corps de fonctions de deux variables (d'apr\`es A. J. de Jong)},
 S\'eminaire Bourbaki,   Exp. No. 949 (2004/2005),   Ast\'erisque  {\bf  307}  (2006), 379--413.

\bibitem{CTGP} J.-L. Colliot-Th\'el\`ene, P. ~Gille and  R. ~Parimala,  {\it
Arithmetic of linear algebraic groups over 2-dimensional geometric fields},  
Duke Math. J.  {\bf 121}  (2004),  285--341.  


\bibitem{CTO} J.--L. Colliot--Th\'el\`ene, M. Ojanguren, {\it Espaces
principaux homog\`enes localement triviaux}, Inst. Hautes \'Etudes
Sci. Publ. Math. {\bf 75} (1992), 97--122.


\bibitem{CTOP} J.--L. Colliot--Th\'el\`ene, M. Ojanguren, R. Parimala, {\it Quadratic forms over fraction fields of two-dimensional Henselian rings and Brauer groups of related schemes}, 
  Algebra, arithmetic and geometry, Part I, II (Mumbai, 2000),  185--217, Tata Inst. Fund. Res. Stud. Math. {\bf  16} (2002), Bombay.

%\bibitem{CTS1} J.-L. ~Colliot-Th\'el\`ene, J.-J. ~Sansuc, {\it  La $R$-\'equivalence sur les tores}, 
 % Ann. Sci. \'Ecole Norm. Sup. {\bf 10}  (1977),  175--229. 


%\bibitem{CTS2} J.--L. Colliot--Th\'el\`ene, J.--J. Sansuc, {\it Fibr\'es
%quadratiques et composantes connexes r\'eelles}, Math. Annalen {\bf 244} (1979), 105--134.

%\bibitem{CTS3} J.-L. ~Colliot-Th\'el\`ene, J.-J. ~Sansuc, {\it  Principal homogeneous spaces under flasque tori: applications},
%J. Algebra {\bf 106} (1987),  148--205. 



\bibitem{DG} M. Demazure et P. Gabriel, {\it Groupes alg\'ebriques}, Masson (1970).


\bibitem{D} A. Ducros, {\it  Dimension cohomologique et points rationnels sur les courbes},  J. Algebra  {\bf 203}  (1998),   349--354.



\bibitem{FI} R. Fossum, B.  Iversen, {\it  On Picard groups of algebraic fibre spaces},   J. Pure Appl. Algebra {\bf  3}  (1973), 269--280.

 
\bibitem{Ga1} R.S. ~Garibaldi,   {\it The Rost invariant has trivial kernel for quasi-split groups of low rank},
Comment. Math. Helv. {\bf  76}  (2001),  684--711. 


\bibitem{Ga2} R.S. ~Garibaldi,   {\it Cohomological invariants : exceptional groups and spin groups}, to appear in Memoirs of the AMS. 


%\bibitem{Ga} R.S. ~GARIBALDI -- Kneser-Tits Conjecture for a rank 1 form of $E_6$ (after
 %Veldkamp), Compositio Math. {\bf 143}  (2007), 191-200. 



\bibitem{GaGi} R.S. ~Garibaldi and P. ~Gille,  {\it Algebraic groups with few subgroups},  preprint (2008).
%to appear in the Proceedings of the London Math. Soc. 


\bibitem{GaH} R.S. ~Garibaldi and D.W. Hoffmann,  {\it Totaro's question on zero-cycles on $G\sb 2$, $F\sb 4$ and $E\sb 6$ torsors},
  J. London Math. Soc. {\bf  73}  (2006),  325--338. 

%\bibitem{GaP} R.S. ~GARIBALDI et  H.-P. ~PETERSSON --
%Groups of outer type $E_6$ with trivial Tits algebras, Transformation Groups %{\bf 12} (2007), 443--474. 


\bibitem{GMS} R.S. Garibaldi, A.A. Merkurjev
and J.-P. Serre, {\it Cohomological invariants in Galois
   cohomology},  University Lecture Series, 28 (2003).
American Mathematical Society, Providence.

\bibitem{GTW} R.S. Garibaldi, J.-P.  Tignol, A.  Wadsworth, {\it  Galois cohomology of special orthogonal groups},   Manuscripta Math. {\bf  93}  (1997),   247--266. 



\bibitem{Gi97} P. Gille,    {\it La $R$-\'equivalence sur les groupes alg\'ebriques r\'eductifs d\'efinis
 sur un corps global},  Inst. Hautes \'Etudes Sci. Publ. Math.  {\bf  86}  (1997), 199--235. 


\bibitem{Gi00} P. Gille,    {\it  Invariants cohomologiques de Rost en caract\'eristique positive},  $K$-Theory 
{\bf  21}  (2000), 57--100.


\bibitem{Gi01} P. Gille,    {\it
Cohomologie galoisienne des groupes quasi-d\'eploy\'es sur des corps de dimension cohomologique $\leq 2$},
 Compositio Math. {\bf 125} (2001), 283--325.

\bibitem{Gi02} P. Gille,    {\it
Unipotent subgroups of reductive groups in characteristic $p>0$},  Duke Math. J.  {\bf 114}  (2002),  307--328.

\bibitem{GiGT} P. Gille,    {\it
An invariant of elements of finite order in semisimple simply  connected algebraic groups}, Journal of Group Theory {\bf 5} (2002),  177-197.


%\bibitem{Gi3} P. ~Gille,  {\it   Lectures at the Morningside Center of Mathematics (Beijing)}, March 2007. 


\bibitem{Gi07} P. Gille,    {\it La conjecture de Kneser-Tits},  s\'eminaire Bourbaki 983 (2007), to appear in Ast\'erisque.

\bibitem{GP07} P. Gille, A.  Pianzola, {\it Galois cohomology 
and forms of algebras over Laurent polynomial rings}, Math. Annalen
{\bf 338}  497--543 (2007)


\bibitem{GiS} P. ~Gille and T. ~Szamuely,  {\it   Central simple algebras and Galois cohomology},
 Cambridge Studies in Advanced Mathematics {\bf  101} (2006), Cambridge University Press.


\bibitem{Gir} J. Giraud, {\it Cohomologie non ab\'elienne},  Die Grundlehren der mathematischen Wissenschaften {\bf 179} (1971), Springer-Verlag. 


\bibitem{GGHZ} D. Goldstein, R.M.  Guralnick, E. W.  Howe, M.E.  Zieve, {\it  Nonisomorphic curves
 that become isomorphic over extensions of coprime degrees},   J. Algebra  {\bf 320}  (2008),   2526--2558. 


%\bibitem{H} G. ~Harder, {\it Halbeinfache Gruppenschemata \"uber Dedekindringen},  Invent. Math. {\bf 4} (1967), 165--191. 



\bibitem{Harder} G. Harder,   {\it  \"Uber die Galoiskohomologie
halbeinfacher Matrizengruppen \un},  Math. Zeit. {\bf 90} (1965), 404-428,
\deux\enskip Math. Zeit. {\bf 92} (1966),  396--415, \trois\enskip J. f\"ur die reine
angew. Math. {\bf 274/5} (1975), 125-138.



\bibitem{I} O. Izhboldin, {\it On the Cohomology
Groups of the Field of Rational Functions},  Coll. Math. St-Petersburg, Ser. 2, AMS Transl.  {\bf 174} (1996),
 21--44.


\bibitem{dJ}  A. J. de Jong,  {\it   The period-index problem for the Brauer group of an algebraic surface}, 
Duke Mathematical Journal {\bf  123} (2004), 71--94.

\bibitem{dJHS}  A. J. de Jong, X. He, J.M. Starr, {\it Families of rationally simply connected
varieties over surfaces and torsors for semisimple groups},
preprint (2008). 



\bibitem{K} K.  Kato,  {\it Galois
cohomology of complete discrete valuation fields},  Lect. Notes in
Math.  {\bf 967} (1982), 215--238.


\bibitem{Kn} M. ~Kneser, {\it 
Lectures on Galois cohomology},
Tata institute (1969). 


%\bibitem{Kn} M. ~Kneser, {\it 
%Starke Approximation in algebraischen Gruppen. I},
%J. reine Angew. Math. {\bf 218} (1965) 190--203. 


%\bibitem{K2} M. A. Knus, {\it Quadratic and hermitian forms over
%rings}, Grundlehren der mat. Wissenschaften {\bf 294} (1991), Springer.

\bibitem{KMRT} M.-A. ~Knus, A. ~Merkurjev, M. ~Rost and J.-P. ~Tignol,  {\it
The book of involutions, avec une pr\'eface de  J. Tits},  American Mathematical Society Colloquium Publications {\bf  44} (1998), American Mathematical Society.


%\bibitem{KP} A. Kulsreshta, R. Parimala,  {\it $R$-equivalence in adjoint classical groups for fields
%of virtual cohomological dimension $2$},  Transactions of the AMS {\bf 360} (2008), 1193--1221. 


%\bibitem{Ga} P. Garrett, {\it Buildings and classical groups},  Chapman
 % and Hall, London, 1997. 
\bibitem{L} M. Lieblich, {\it Deformation theory and rational points on rationally
connected varieties}, preprint (2008).


\bibitem{LT} Lewis, J.-P. Tignol, {\it Classification theorems for central simple
algebras with involution}, Manuscripta Math. {\bf 100}, (1999), 259--276.

\bibitem{Ma} B. Margaux, {\it Passage to the limit in non-abelian \v Cech cohomology}, Journal of Lie Theory {\bf 17} (2007), 591--596.

%\bibitem{M} J.S. Milne, { \it Etale cohomology}, Princeton University Press (1980).


\bibitem{M0} A. S. ~Merkurjev,  {\it Simple algebras and quadratic forms}, Izv. Akad. Nauk SSSR {\bf 55} (1991), 
218--224.


\bibitem{M1} A. S. ~Merkurjev,  {\it $K$-theory of simple algebras}, 
 $K$-theory and algebraic geometry: connections with quadratic forms and division algebras (Santa Barbara, CA, 1992),  65--83, Proc. Sympos. Pure Math. {\bf  58.1}
(1995), Amer. Math. Soc.

\bibitem{M3} A. S. ~Merkurjev,  {\it  $K$-theory and algebraic groups}, 
 European Congress of Mathematics, Vol. II (Budapest, 1996),  43--72, Progr. Math. {\bf  169} (1998), Birkh\"auser.

%\bibitem{M4} A. S.  ~Merkurjev, {\it Invariants of algebraic groups},  J. reine angew. math.  {\bf 508}  (1999), 127--156.

%\bibitem{M5}  A. S. ~Merkurjev,  {\it The group $SK\sb 1$ for simple algebras}, 
%$K$-Theory {\bf 37} (2006), 311--319. 


%\bibitem{MPW} A.  S. ~Merkurjev , I.  ~Panin and A. R. ~ Wadsworth, {\it 
 %Index reduction formulas for twisted flag varieties. II},  $K$-Theory  {\bf 14}  (1998),   101--196. 


 \bibitem{MS} A.--S. ~Merkurjev and A.--A. ~Suslin, {\it $\cal K$-cohomology of Severi-Brauer varieties and
 norm residue  homomorphism}, Izv. Akad. Nauk SSSR
{\bf  46}  (1982), 1011--1046, english translation: Math. USSR Izv. {\bf 21} (1983),
307--340.



 \bibitem{MT} A.--S. ~Merkurjev and  J.-P. ~Tignol, {\it  The multipliers of similitudes and the Brauer group of homogeneous varieties}, 
  J. Reine Angew. Math.  {\bf 461}  (1995), 13--47. 



\bibitem{MV} F.  ~Morel et V. ~Voevodsky,  ${\bf A}^1$-homotopy theory of schemes, 
 Inst. Hautes \'Etudes Sci. Publ. Math.   {\bf 90}  (1999), 45--143.



 \bibitem{Pa} R.  ~Parimala, {\it Homogeneous varieties---zero-cycles of degree one versus rational points}, 
  Asian J. Math.  {\bf 9}  (2005),   251--256. 

\bibitem{PR} V. P. ~Platonov and A. ~Rapinchuk,  {\it Algebraic groups and number theory},
 Pure and Applied Mathematics {\bf  139} (1994), Academic Press.


 %\bibitem{N} H. Nagao, {\it  On ${\rm GL}(2,\,K[x])$},   J. Inst. Polytech. Osaka City Univ. Ser. A  {\bf 10} (1959), 117--121.


%\bibitem{Ono} T. Ono, {\it On the Arithmetic of Algebraic Tori}, 
%Annals of Math. {\bf 74} (1961),  101-139.

%\bibitem{Ra}  M. S. Raghunathan, {\it Principal bundles
% admitting a rational section}, Invent. math. {\bf 116} (1994), 409-423. 


%\bibitem{RR} M. S. Raghunathan, A.  Ramanathan, {\it Principal bundles on the %affine line},
 % Proc. Indian Acad. Sci. Math. Sci.  {\bf 93}  (1984),   137--145.


%\bibitem{Sa} D. J. ~Saltman,  {\it Retract rational fields and cyclic Galois extensions},  Israel J. Math. 
 %{\bf 47} (1984), 165--215.


\bibitem{Sa} J.-J. Sansuc, {\it  Groupe de Brauer et arithm\'etique des groupes alg\'ebriques lin\'eaires},
 J. reine angew. Math. {\bf 327} (1981), 12--80.

%\bibitem{Sc} W.  ~Scharlau,  {\it  Quadratic and Hermitian forms}, 
% Grundlehren der Mathematischen Wissenschaften {\bf 270} (1985), Springer-Verlag.


\bibitem{Se0} J-P. ~Serre,  {\it  Cohomologie galoisienne des groupes alg\'ebriques lin\'eaires},
Colloque sur la th\'eorie des groupes alg\'ebriques lin\'eaires, Bruxelles (1962), 53--68.


\bibitem{Se1} J-P. ~Serre,  {\it  Cohomologie galoisienne},
cinqui\`eme \'edition r\'evis\'ee et compl\'et\'ee, Lecture Notes in
Math.  {\bf 5}, Springer-Verlag.

\bibitem{Se2} J-P. ~Serre,  {\it Cohomologie galoisienne:
Progr\`es et probl\`emes}, S\'eminaire Bourbaki, expos\'e 783
(1993-94), Ast\'erisque {\bf 227} (1995).


\bibitem{SGA3} {\it S\'eminaire de G\'eom\'etrie alg\'ebrique de
l'I.H.E.S., 1963-1964, sch\'emas en groupes, dirig\'e par M.
Demazure et A. Grothendieck},  Lecture Notes in Math. 151-153.
Springer (1970).


%\bibitem{Se1}  J.-P. Serre, {\it
%Arbres, amalgames, ${\rm SL}\sb{2}$},  r\'edig\'e avec la collaboration de Hyman Bass, Ast\'erisque {\bf  46} (1977).


\bibitem{Sh}  S. Shatz, {\it  Cohomology of artinian group schemes over local fields}, 
  Ann. of Math. {\bf   79} (1964),  411--449.

\bibitem{Sm} N. Semenov, {\it Motivic construction of cohomological invariants}, preprint (2008).

\bibitem{Starr} J. Starr, {\it Rational points of rationally connected and 
rationally simply connected varieties}, preprint (2008). 

%\bibitem{Sp} T. A. Springer, {\it  Reductive groups},  Automorphic forms, representations and $L$-functions (Proc. Sympos. Pure Math., Oregon State Univ., Corvallis, Ore., 1977), Part 1,  pp. 3--27, Proc. Sympos. Pure Math., XXXIII, Amer. Math. Soc., Providence, R.I., 1979.

%\bibitem{S1} C. Soul\'e, {\it   Groupes op\'erant sur un complexe simplicial avec domaine fondamental},
 % C. R. Acad. Sci. Paris S\'er. A-B  {\bf 276}  (1973), 607--609.

%\bibitem{S2} C. Soul\'e, {\it  Chevalley groups over polynomial rings},  Homological group theory (Proc. Sympos., Durham, 1977),  %359--367, London Math. Soc. Lecture Note Ser. {\bf  36} (1979), Cambridge Univ. Press.



%\bibitem{ST}  T. A. ~Springer and  R. ~Steinberg,  {\it  Conjugacy classes}, 
 %  Seminar on Algebraic Groups and Related Finite Groups (The Institute for Advanced Study, Princeton, N.J., 1968/69),   Lecture Notes in Math.  {\bf 131} (1970), 167--266,Springer. 


\bibitem{St}   R. ~Steinberg, {\it Regular elements 
of semisimple algebraic groups},
Pub. Math. IHES {\bf 25} (1965), 281--312.


%\bibitem{St2}   R. ~Steinberg,  {\it  Lectures on Chevalley groups}, notes prepared by John Faulkner and Robert Wilson, Yale University (1968), New Haven. 

\bibitem{Su}  A. A. ~Suslin, {\it  Algebraic $K$-theory and
 the norm--residue homomorphism}, J.~Soviet. {\bf 30} (1985), 2556--2611.


\bibitem{Sw} R. Switzer, {\it  Algebraic topology---homotopy and homology},  Die Grundlehren der mathematischen Wissenschaften, Band 212 (1975), Springer-Verlag.

%\bibitem{T64} J. Tits,  {\it Algebraic and abstract simple groups},  Ann. of Math. {\bf   80} (1964), 313--329.


\bibitem{T66} J.~Tits, {\it   Classification of algebraic semisimple groups},     
Algebraic Groups and Discontinuous Subgroups,  Proc. Sympos. Pure Math., Boulder, Colo.  (1965)  pp. 33--62,
 Amer. Math. Soc.


\bibitem{T71} J. Tits, {\it Repr\'esentations lin\'eaires irr\'eductibles 
d'un groupe r\'eductif sur un corps quelconque},  J. reine ang. Math. {\bf 247} (1971),  196-220.


%\bibitem{T73} J. Tits,  {\it Buidings of spherical type and finite BN-pairs},
%Lect. Notes in Math. {\bf   396} (1973), Springer.

%\bibitem{T76} J. Tits,Groupes de Whitehead de groupes alg\'ebriques simples sur un corps 
%(d'apr\`es V. P. Platonov et al.),    S\'eminaire Bourbaki,
%29e ann\'ee (1976/77),  Exp. No. 505, pp. 218--236,
% Lecture Notes in Math. {\bf  677} (1978).

%\bibitem{T90} J. Tits, {\it Strongly inner anisotropic forms of simple algebraic groups},
%J. Algebra {\bf 131} (1990), 648--677. 


\bibitem{T90} J. Tits, {\it  R\'esum\'e des cours du Coll\`ege de France
1990--91}, Annuaire du Coll\`ege de France.


\bibitem{T92} J. Tits, {\it Sur les degr\'es des extensions de corps d\'eployant les groupes
alg\'ebriques simples}, C. R. Acad. Sci. Paris S{\'e}r. I Math. \textbf{315}
(1992),  1131--1138.


\bibitem{To} B. Totaro, {\it Splitting fields for $E\sb 8$-torsors},  Duke Math. J.  {\bf 121}  (2004),   425--455. 

 
\bibitem{W}  A. Weil, {\it  Algebras with involutions and the classical groups},  J. Indian Math. Soc. {\bf  24} (1960),  589--623.

%\bibitem{Wg} S. ~Wang, {\it  On the commutator group of a simple algebra},
%Amer. J. Math. {\bf 72} (1950), 323--334. 

\end{thebibliography}



\title{Serre's conjecture II: a survey}

\author{Philippe Gille}


\date{}
\smallskip

 \maketitle

\bigskip

%\thanks{{\tt Version of \today.}}





\bigskip

 \begin{small}
\noindent {\bf Abstract\footnote{{\tt  Version of \today.}}:} Our goal is to provide an up to date survey of Serre's conjecture II (1962) on the vanishing of
Galois cohomology for simply connected semisimple groups defined over a field of 
cohomological dimension $\leq 2$. 
\bigskip

\noindent {\bf Keywords:} Galois cohomology, linear algebraic groups.

\noindent {\bf MSC: 20G05}


%\noindent
%{\bf Keywords: }

\end{small}



%{\small \tableofcontents }


 \vskip14mm 



%\date{December 2008}



%\vskip14mm


\section{Introduction}


 Serre's original conjecture II (1962) states that the Galois cohomology set $H^1(k,G)$
vanishes for a semisimple simply connected  algebraic group $G$ defined over a perfect field $k$ of cohomological dimension 
$\cd(k) \leq 2$ \cite[\S 4.1]{Se0} \cite[II.3.1]{Se1}.
In other words,  that all $G$-torsors (or principal homogeneous spaces) over $\Spec(k)$ are trivial.
%Possibly, this conjecture has been done with analogy with the following result of Topology (??).

For example, if $A$ is a central simple algebra defined over a field $k$ and $c \in k^\times$,
the subvariety
$$
X_c:=\{ \nrd(y)=c \} \, \subset \GL_1(A)  
$$
of elements of reduced norm $c$ is a torsor under the special linear group
$G=\SL_1(A)$ which is semisimple and simply connected.
If $\cd(k) \leq 2$, we expect then that this $G$-torsor is trivial, i.e.
 $X_c(k) \not= \emptyset$. By considering all scalars $c$, we expect then that the reduced norm map
$A^\times \to k^\times$ is surjective.


For imaginary number fields, the surjectivity of the reduced norm map goes back to Eichler in 1938 (see \cite[\S 5.4]{Kn}).
For function fields of complex surfaces, this follows from the Tsen-Lang theorem given that
the reduced norm is a homogeneous form of degree $\deg(A)$ in $\deg(A)^2$-indeterminates
\cite[II.4.5]{Se1}.
The general case of the surjectivity of  reduced norm maps
was established in 1981 by  Merkurjev and Suslin  \cite[th. 24.8]{Su}. As we shall explain below, this fact essentially 
characterizes fields of cohomological dimension $\leq 2$.


Throughout its history, the evidence for and progress towards establishing conjecture II has been gathered by either considering special classes of fields, or by looking at the implications that the conjecture would have on the classification of algebraic groups. We will explore both points of view in this survey. 

>From the groups point of view, the strongest evidence
for the validity of the conjecture is given by the description of the classical groups  established in 1995 by Bayer  and Parimala  \cite{BP}.
>From the point of view of fields,  we know that the  conjecture holds
in the case of imaginary number fields (Kneser \cite{Kn}, Harder \cite{Harder}, 
Chernousov \cite{C89}, see \cite[\S 6]{PR}),
and  more recently for function fields of complex surfaces.
For exceptional groups with no factors of type $E_8$,
the relevant reasonings and references are given in \cite{CTGP}. A general proof for all types using deformation methods was given in 2008 by  He-de Jong-Starr \cite{dJHS}.
This  result has a clear geometric meaning:
If $G/\C$ is a semisimple simply connected group and $X$ a smooth complex surface, then 
a  $G$-torsor over $X$ (or a $G$-bundle) is  locally trivial with respect to the Zariski topology (see \S \ref{cor-main}).


There are previous surveys on Galois cohomology discussing Serre's conjecture II and related topics.
Tits' lectures at Coll\`ege de France from 90-91 discuss 
the Hasse principle  and group classification \cite{T90}. Serre's Bourbaki seminar  deals among other 
things with  
progress and the status of  conjecture II as of  1994. For classical groups,  there is 
 Bayer's survey  \cite{Ba}.
For function fields of surfaces, see the surveys of 
Starr \cite{Starr} and Lieblich \cite{L}.

For exceptional groups (trialitarian, type $E_6$, $E_7$ and
$E_8$), the general conjecture is still open in spite of some considerable progress
\cite{C03}\cite{CTGP}\cite{CTOP}\cite{Gi01}. 



We finish the introduction by mentioning that Serre's conjecture II can be linked  with  analogous considerations in Topology
within Morel-Voevodky's theory \cite{MV}. Indeed, if $G$ is a semisimple simply connected complex group,
we know that $\pi_1( G)= \pi_2(G)=0$, hence $G$ is $2$--connected.
 Then for every $CW$-complex of dimension $\leq 2$,
the $G$-bundles over $X$ are trivial (cf. \cite[th. 11.34]{Sw}).

%\footnote{The same theme is found in the so-called Serre's conjecture:   Projective modules of finite type over a polynomial ring with coefficients in a field are free. This assertion, which was established independently by Quillen and Suslin, serves as the algebraic counterpart to the trivial topological fact that all vector bundles over (real or complex) affine space are trivial.} 


\bigskip

\noindent{\bf Acknowledgement.}
We thank Vladimir Chernousov, Skip Garibaldi, Fabien Morel, Arturo Pianzola and Jean-Pierre Tignol 
for useful comments which improved the exposition.

\medskip



\section{Fields of cohomological dimension $\leq 2$}

Let $k$ be a field and $l$ be a prime.
Recall that $k$ is of $l$--cohomological dimension at most $d$, written  $\cd_l(k) \leq d,$ if 
$H^i(k,A) =0$ for every finite $l$--primary Galois  module $A$ and for all  $i \geq d+1$.
We know that this assertion is equivalent to the vanishing of
$H^{d+1}(L, \Z/l\Z)$ for any  finite separable extension $L/k$.
Recall the following examples of fields of cohomological dimension $2$. 

\begin{examples} {\rm
 (1) Imaginary number fields;

\smallskip

\noindent (2) Function fields of complex surfaces;

\smallskip

\noindent (3)
 Merkurjev's tower of fields $F_\infty$, namely an extension 
of $\C(X_1,...,X_{2n})$ such that the $u$-invariant is $u(F_\infty)=2n$.
This means that  every $2n+1$-dimensional quadratic form over $F_\infty$
is isotropic but the form $\langle X_1,X_2, \cdots X_{2n}\rangle$ remains anisotropic over $F_\infty$.
Furthermore the tensor product of the quaternion algebras $(X_{2i-1}, X_{2i})$ for $i=1,...,n$
  is a division algebra over $F_\infty$ \cite{M0}\cite[th. 3]{M1}. 
}
\end{examples}

The third example shows that  central simple algebras and quadratic forms 
are not in general low dimensional objects. 
We have already mentioned the following characterization which uses 
 Merkurjev-Suslin's theorem \cite{MS}.


\begin{theorem}\label{MSjo} \cite[th. 24.8]{Su}
Let $l$ be an invertible prime in $k$.
The following are equivalent:

\begin{enumerate}
\item $\cd_l(k) \leq 2$.
\item For any finite separable extension $L/k$ and any $l$-primary central simple $L$--algebra
$A/L$, the reduced norm $\nrd: A^\times \to L^\times$ is surjective.
\item For any finite extension $L/k$ and any $l$-primary central simple $L$--algebra
$A/L$, the reduced norm $\nrd: A^\times \to L^\times$ is surjective.
\end{enumerate}
 
\end{theorem}

We have  added here the easy implication   $2) \Longrightarrow 3)$ which follows of 
by a familiar transfer argument.
We say that $k$ is of cohomological dimension  $\leq d$ if 
$k$ is of $l$--cohomological dimension $\cd_l(k) \leq d$ for all primes $l$.

If $k$ is of positive characteristic $p$, we always have $\cd_p(k) \leq 1$; this  explains the necessary change
in  the following analogous statement.


\begin{theorem}\label{MSp} \cite[th. 7]{Gi00}
Assume that ${\rm char}(k) =p >0$. The following are equivalent:

\begin{enumerate}
\item $H^3_{p}(L)=0$ for any finite separable extension $L/k$;
\item For any finite separable extension $L/k$ and any $l$-primary central
 simple $L$--algebra $A/L$, the reduced norm $\nrd: A^\times \to L^\times$ is surjective.
\end{enumerate}
 
\end{theorem}

Here  $H^3_{p}(k)$ is the cohomology group of Kato 
defined by means of logarithmic differential forms \cite{K}, see also \cite[\S 9]{GiS}.
We shall say that  $k$ is of separable $p$--dimension $\leq d$ if
$H^{d+1}_{p}(L)=0$ for all finite separable extension $L/k.$
This defines in the obvious way the separable dimension $\sd_p(k)$
of $k$. For $l \not=p$, we define 
 $\sd_l(k)=\cd_l(k)$.\footnote{Kato defined the $p$--dimension $\dim_p(k)$  as follows \cite{K}.
If $[k:k^p] = \infty$, define  $\dim_p(k)=\infty$.
If  $[k:k^p] = p^r < \infty$, 
  $\dim_p(k)=r$ if  $H^{r+1}_{p}(L)=0$ for  any finite  extension $L/k$,
and   $\dim_p(k)=r+1$ otherwise.}
If $k$ is perfect, then $H^i_p(L)=0$ for every finite extension $L/k$ and for every  $i \geq 2$.
Hence if $k$ is perfect and of cohomological dimension $\leq 2$,
$k$ is of separable  dimension $\leq 2$.



\begin{examples}{\rm
\noindent (1) The function field of a curve over a finite field is of separable dimension $2$.

\noindent (2) The function  field $k_0(S)$ of
a surface over an algebraically closed field $k_0$ of characteristic $p\geq 0$ 
is of separable dimension $2$.

\noindent 
(3) Given an arbitrary field $F$, Theorems \ref{MSjo} and \ref{MSp} provide a way to construct 
a ``generic'' field extension $E/F$ of separable dimension $2$, see Ducros \cite{D}.
}
\end{examples}



We can now state the strong form of Serre's conjecture II.
For each simply connected group $G$, Serre defined the set $S(G)$ of  primes in terms of
the Cartan-Killing type of $G$, cf.~\cite[\S 2.2]{Se2}. 
For  absolutely simple almost groups, the sets $S(G)$ are as follows.



\begin{table}[ht]
\centering
\caption{$S(G)$ for absolutely almost simple groups} \label{torsion.table}
\bigskip

\begin{tabular}{|c|c|} \hline
type&$S(G)$\\ \hline
$A_n$ ($n \ge 1$) & $2$ and the prime divisors on $n+1$\\
$B_n$ ($n \ge 3$),  $C_n$ ($n \ge 2$), $D_n$ (non trialitarian for $n = 4$)&2 \\
$G_2$&2 \\
trialitarian $D_4, F_4, E_6, E_7$&2, 3\\
$E_8$&2, 3, 5 \\ \hline
\end{tabular}

\end{table}



\begin{conjecture}\label{strong} 
 Let $G$ be a semisimple simply connected algebraic group.
Assume that $\sd_l(k) \leq 2$ for every prime $l \in S(G)$, then
$H^1(k,G)=0$.
\end{conjecture}



In the  original conjecture, $k$ was assumed perfect and of cohomological dimension $\leq 2$.
In characteristic $p>0$, Serre's strenghthened  question assumed furthermore that $[k:k^p] \leq p^2$
if $p$  belongs to  $S(G)$ 
\cite[\S 5.5]{Se2}.  In view of all known results, it would seem that there is no need for this assumption.


 
Conjecture \ref{strong} is indeed stronger than the original one. 
Theorems \ref{MSjo} and \ref{MSp} show that this strong version of the conjecture holds for groups of inner type $A,$ and that
 the hypothesis on $k$
are sharp.



\section{Link between the conjecture and  the classification of groups}

The classification of semisimple groups reduces essentially to that of
semisimple simply connected groups $G$ which are absolutely almost simple  \cite[\S 31.5]{KMRT}\cite{T66}.
This means that $G \times_k k_s$ is isomorphic to $\SL_{n,k_s}$, $\Spin_{2n+1, k_s}$, $\Sp_{2n, k_s}$,   $\Spin_{2n, k_s}$,...

Let $G/k$ be such a $k$--group and
let $G \to G_{ad}$ be the adjoint quotient of $G$.
Denote by  $G^{q}$ its quasi-split form and
by  $G^{q}_{ad}$ its adjoint quotient.
Then $G$ is  an inner twist of $G^{q}$, i.e. there exists a cocycle 
 $z \in  Z^1(k_s/k,G^q_{ad}(k_s))$ such that  $G \cong {_zG}^q$.
We  identify then  $G$ and ${_zG}^q$.

Converely, we know that there exists a unique class $\nu_G=[a] \in H^1(k,G_{ad})$
such that  $G^q \cong {_aG}$  \cite[31.6]{KMRT}.
We denote by $z^{op} \in  Z^1(k,{_zG_{ad}^q})$ the opposite cocycle of $z$, 
it is defined by $\sigma \mapsto z_\sigma^{-1} \in {_zG}(k_s)$.

We have $G^q \cong {_{z^{op}} \bigl({_zG}^q}\bigr)$.
Hence the image of $\nu_G$ under 
$H^1(k,G_{ad}) \simlgr   H^1(k, {_zG^q}_{ad})$ is nothing but $[z^{op}]$. 
We have  an exact sequence
$$
1 \to Z(G) \to G \to G_{ad} \to 1
$$
of $k$--algebraic groups with respect to the $fppf$-topology (faithfully flat of finite presentation, see \cite[III]{DG}
or \cite{Sh}).
This gives rise to an exact sequence of pointed sets \cite[app. B]{BFT}
$$
1 \to Z(G)(k) \to G(k) \to G_{ad}(k) \buildrel \varphi_G \over \lgr 
$$
$$
H^1_{fppf}(k,Z(G)) \to H^1_{fppf}(k,G)  \to H^1_{fppf}(k,G_{ad}) \buildrel \delta_G  \over \lgr  H^2_{fppf}(k,Z(G)).
$$
The homomorphism  $\varphi_G$ is called the characteristic map and
the mapping $\delta_G$ is the boundary. Since $G$  (resp. $G_{ad}$) are smooth,
the $fppf$-cohomology of $G$ (resp. $G_{ad}$) coincide with Galois cohomology \cite[XXIV.8]{SGA3}, 
i.e. we have a bijection  $H^1(k,G)  \simlgr  H^1_{fppf}(k,G)$.
Following  \cite[31.6]{KMRT}, one  defines  the Tits class $T_G$ of $G$ by the formula
$$
t_G= \, - \, \delta_G\bigl(   \nu_G  \bigr) \in  H^2_{fppf}(k,Z(G)).
$$
By the compatibility property under
 the torsion bijection  $\tau_z$ \cite[IV.4.2]{Gir}\footnote{Note  that $Z(G)=Z(G^{q})$ since 
$G^q_{ad}$ acts trivially on $Z(G^{q})$.} 
$$
\begin{CD}
H^1(k,G_{ad}) @>{\delta_G}>> H^2_{fppf}(k,Z(G)) \\
@V{\tau_z}V{\wr}V @V{ ? \, + \,  \delta_{G^q}([z])}V{\wr}V \\
H^1(k,G^q_{ad}) @>{\delta_{G_{ad}}}>> H^2_{fppf}(k,Z(G^q)) ,\\
\end{CD}
$$
we see that $t_G= \delta_{G^q}( [z])$ which is indeed Tits definition \cite[\S 1]{T90}.
%Recall that a strongly inner form of $G$ is 
%the twist of $G$ by a cocycle $a \in Z^1(k_s/k, G(k_s))$ \cite{T90}. 


\begin{proposition} 
Assume that $H^1(k,G)=1$.

\begin{enumerate}
\item The boundary map 
$H^1(k,G_{ad}) \to H^2_{fppf}(k, Z(G))$ has trivial kernel.

\item Let $G'$ be an inner  $k$--form of $G^{q}$.
Then $G$ and $G'$ are isomorphic if and only if $t_G=t_{G'}$. 

\end{enumerate}
\end{proposition}


\begin{proof}
(1) This follows  from the exact sequence above.

\smallskip

\noindent (2) Let $z' \in Z^1(k,G^q_{ad})$ be a cocycle such that
$G' \cong {_{z'}G}$. We assume that  $t_G=t_{G'}$.
Hence $\delta_{G^q}( [z])= \delta_{G^q}( [z']) \in H^2_{fppf}(k,Z(G^q))$.
The compatibility above shows that
$$
\tau_z^{-1}([z']) \in \ker\bigl( H^1(k,G_{ad}) \bigr)\to H^2_{fppf}(k,Z(G)) \bigr).
$$
By 1), we have $\tau_z^{-1}([z'])= [1] \in H^1(k,G_{ad})$, hence
$[z]=[z'] \in H^1(k,G_{ad})$.
Thus $G$ and $G'$ are $k$--isomorphic.
\end{proof}



Summarizing then, the validity of Serre's conjecture II implies that semisimple $k$-groups 
are classified by their quasi-split forms and their Tits classes.
For more precise results for classical groups, see Tignol-Lewis \cite{LT}.
The classification  is of special importance in view of the rationality question for  groups
(Chernousov-Platonov \cite{CP}, see also Merkurjev \cite{M3}), and consequently also 
 for the  Kneser-Tits problem (Gille, \cite{Gi07}).




\section{Approaches to  the conjecture} 

We would like to describe some of the methods that have been used to attack the conjecture to date, 
and their limitations. We should point out that separating each of the methods and looking at them individually is a bit artificial. In practice, most work is carried by simultaneously combining the different methods.



\subsection{Subgroup trick}
 
We illustrate how this method works by using  the following example due to Tits \cite{T92}. Let $G/k$ be the
split semisimple simply connected group of type  
$$
\mbox{$E_6 \qquad$}
\begin{picture}(65,45)
\put(00,02){\line(1,0){80}}
\put(40,02){\line(0,1){20}}
\put(00,02){\circle*{3}}
\put(20,02){\circle*{3}}
\put(40,02){\circle*{3}}
\put(60,02){\circle*{3}}
\put(80,02){\circle*{3}}
\put(40,22){\circle*{3}}
\put(-5,-11){$\alpha_1$}
\put(15,-11){$\alpha_3$}
\put(35,-11){$\alpha_4$}
\put(55,-11){$\alpha_5$}
\put(75,-11){$\alpha_6$}
\put(35,30){$\alpha_2$}
\end{picture}
\mbox{\qquad}
$$

\smallskip

\noindent Assume here that $k$ is infinite. Let $z \in Z^1(k_s/k,G),$ and consider the 
twisted group $G'={_zG}$. Since $t_{G'}=0$, the 27-dimensional standard representation of 
$G$ of highest weight $\overline \omega_6$ descends to $G'$ by \cite{T71}. We have then
a representation $\rho':   G' \to \GL(V)$. The point is that  
$G'$ has a dense orbit in the projective space $X=\P(V)$, so there exists a $k$-rational point $[x]$ 
in this orbit. The connected stabiliser $(G'_x)^0$ is then semisimple of type $F_4$ \cite[9.12]{Ga2}.
Assuming that Conjecture \ref{strong} holds for groups of type $F_4$, it follows that   
$(G'_x)^0$ is split. Hence $G'$ has relative rank $\geq 4$ and a glance on Tits tables \cite{T66}
tells us that $G'$ is split. It is then easy to conclude that $[z]=1 \in H^1(k,G)$.


The subgroup trick, and variations thereof, was fully investigated by Garibaldi in his Lens lectures \cite{Ga2}.
The underlying topic is that of prehomogeneous spaces, namely projective $G$-varieties with a dense orbit. 



Unfortunately, this  trick  works only in few cases.
Tits has shown that the general form of type $E_8$ is ``almost abelian''
namely that it has no non trivial (connected) reductive subgroups other than 
maximal tori  \cite{T92}. Garibaldi and the author have shown that the general trialitarian group
is almost abelian \cite{GaGi}.





% \subsection{Inductive process for quasi-split groups} 


\subsection{Rost invariant}
In this case, the idea is to derive Serre's conjecture II
from a more general setting. The Rost invariant \cite{GMS} generalizes 
the Arason invariant for $3$-fold Pfister form which attaches (in  characteristic $\not =2$) to a Pfister form
$\phi=\langle\langle a,b,c\rangle \rangle$ the cup-product $e_3(\phi)={(a) \cup (b) \cup (c)} 
\in H^3(k, \Z/2\Z)$.  We see it now as the cohomological invariant 
$H^1(k, \Spin_8) \to H^3(k, \Z/2\Z(2))$.
More generally, for $G/k$ simply connected and absolutely almost simple, there is a cohomological invariant
$$
r_k : H^1(k,G) \to H^3(k, \Q/\Z(2))
$$  
where the $p$-primary part has to be understood in  Kato's setting \cite{GMS}.
If this invariant has trivial kernel, then $H^1(k,G)=1$ for 
$G/k$ satisfying the hypothesis of Conjecture \ref{strong}.
This is the case for $\Spin_8$ by Arason's theorem, namely the fact that the invariant $e_3(\phi)$ determines $\phi$.


\subsection{Serre's injectivity question}

A special case of  a question raised by Serre in  1962 (\cite{Se0}, see also  \cite[\S 2.4]{Se2}) is the following.

\begin{question}\label{q-inj} Let $G/k$ be a connected linear algebraic group. Let $(k_i)_{i=1,..,r}$ be
a family of finite field extensions of $k$ such that $g.c.d.( [k_i:k])=1$. 
Is the kernel of  the map
$$
H^1(k,G) \to \prod\limits_{i=1,..,r}  H^1(k_i,G) 
$$
 trivial  ?
\end{question}


\begin{remarks}{\rm

\noindent (1) The hypothesis of  connectedness is necessary since there are counterexamples
with finite constant groups \cite{GGHZ}\cite{Pa}.
Let us mention here another counterexample concocted
by S. Garibaldi. Suppose that  $k$ supports central division algebras $B_i$ of degree $i$ for $i = 3, 5$. 
There are extensions $k_i$ of $k$ of degree $i$ that split $B_i$. The groups $\SL_1(B_3 \otimes_k  B_5 )$ and 
$\SL_1(B^{op}_3 \otimes_k B_5)$ are isomorphic over
$k_i$ for $i = 3, 5$ but they are not $k$-isomorphic.  So Serre's injectivity fails for the group
 $\Aut\bigl(\SL(B_3 \otimes B_5)\bigr)$, which has just two connected components. 
\smallskip


\noindent  (2) Question \ref{q-inj}  has been  generalized by Totaro  \cite[question 0.2]{To}.  See also
\cite{GaH}.
 

\smallskip

\noindent (3) If $k$ is of positive characteristic $p$, there exists a complete DVR $R$
with residue field $k$ and an $R$--group scheme
${\mathfrak G}$ with special fiber $G$ and such that the fraction field $K$ of $R$ is of characteristic zero.
An answer for ${\mathfrak G}_K$ to Serre's question yields 
an answer for $G$. 
A fortiori,  we can assume without lost of generality,  that the extensions $k_i/k$ are separable.
}
\end{remarks}

We shall rephrase  question \ref{q-inj} in terms of special fields.


\begin{definition} Let $l$ be a prime. We say that a field $k$ is $l$--special if
every finite separable extension of $k$ is of degree a power of $l$.
\end{definition}

 
The subfield $k_l$ of $k$ consisting of elements fixed by a $p$-Sylow subgroup of $\Gal(k_s/k)$ is $l$-special.  We call
 $k_l$ a \emph{co-$l$-closure} of $k$.
If we restrict  Serre's question for finite  separable extensions $k_i/k$
and consider all cases, it can be  rephased  by asking  whether the map
$$
H^1(k,G) \to \prod\limits_{l}  H^1(k_l,G) 
$$
has trivial kernel for $l$ running over the primes.
If  the  answer to this  question is in the affirmative, then conjecture II  becomes  a question
for $l$-special fields for primes $l$ in $S(G)$. 

Indeed there are very few cases for which  the answer to Serre's question is known:
unitary groups (Bayer-Lenstra \cite{BL}), 
%few other classical groups \cite{BS}, 
 groups of type $G_2$,
  quasi-split groups of
type  $D_4$, $F_4$, $E_6$, $E_7$  \cite{Gi97} \cite{C03} \cite{Ga1}.   


If we know that the Rost invariant has trivial kernel, then we easily deduce that the answer to Question 
\ref{q-inj} is yes. Thus we can answer Serre's question for groups of type $G_2$,
and quasi-split semisimple simply connected  groups of type
$D_4$, $F_4$, $E_6$ and $E_7$.



\section{Known cases in terms of groups}

\subsection{Classical groups}

Recall that a semisimple simply connected  group is classical if 
its factors are of type $A$, $B$, $C$ or  $D,$ and
there is no triality involved. 

\begin{theorem} \label{classic}
Let $G$ be a semisimple simply connected classical group 
which is absolutely almost simple.
 Then $H^1(k,G)=1$.
\end{theorem}

If $k$ is perfect or ${\rm char}(k) \not =2$, this is  the original Serre's conjecture II proven
 by Bayer-Parimala \cite{BP}.
The general case  is done in recent work by 
Berhuy-Frings-Tignol \cite{BFT}. Their proof is based on 
Weil's  presentation of classical  group in terms of unitary groups of  algebras with involutions \cite{W},
This proof  is characteristic free.  It provides  quite a different  approach to the conjecture than that present in  Bayer and Parimala's  work.

Possibly the most tricky case is that of outer groups of type $A$, namely
unitary groups of central simple algebras equipped with an involution 
of the second kind. It is enough to think  about the  number field case
using Landherr's theorem \cite[\S 5.5]{Kn} to see how complicated the case these outer groups is.  


\subsection{Quasi-split exceptional  groups}

For this type of groups, the best results to date have been obtained by investigating the Rost invariant.


\begin{theorem}\label{rost}  
Let $G/k$ be a quasi-split semisimple simply connected group of Cartan-Killing type $G_2$, $F_4$, $D_4$, $E_6$ or $E_7$. 
 Then the Rost invariant $H^1(k,G) \to H^3(k,\Q/\Z(2))$ has trivial kernel.
\end{theorem}


Note first that this kind of statement reduces to the characterictic zero case by a lifting argument \cite{Gi00}.
For the cases $G_2$ and $F_4$, see \cite{BP} or \cite{Se2}.
As pointed out by Garibaldi,  the $D_4$ case is done in the Book of Involutions but
not stated in this form. We need to know that a trialitarian algebra whose underlying  algebra
is split  arises  as the endomorphism of a twisted composition \cite[44.16]{KMRT}
and to use results on degree 3 invariants of twisted compositions ({\it ibid}, 40.16).
For type $E_6$ and $E_7$,  this is due independently to  Chernousov \cite{C03} and Garibadi \cite{Ga1}.

Thus Conjecture \ref{strong} holds for quasi-split groups of all types other than $E_8.$
The author has given  an independent proof  based on
Bruhat-Tits theory which is  quite different in spirit from the one outlined above \cite{Gi01}. 
For the split group of type  $E_8,$ which will be denoted by $E_8$, the Rost invariant has  in general a non trivial kernel 
(for the field of real numbers and also for suitable fields of
cohomological dimension $4$, see \cite[appendix]{GiGT}). In  characteristic $0$,
Semenov constructed recently a higher  invariant
$$
\ker\bigl[ H^1(k,E_8) \to H^3(k,\Q/\Z(2)) \bigr] \to H^5(k,\Z/2\Z)  
$$
which is non trivial since it is does not vanish for the field of real numbers \cite[\S 8]{Sm}.
Moreover, Semenov's invariant has trivial kernel for $2$-special fields.

By means of norm group  of varieties of Borel subgroups, the case of quasi-split groups is the
key ingredient  for proving the following.

\begin{theorem}\label{center} \cite[th. 6]{Gi01}
Let $G/k$ be a semisimple simply connected group which satisfies the hypothesis of Conjecture  \ref{strong}. 
Let $\mu \subset G$ be a finite central subgroup of $G$. Then the characteristic map
$$
(G/ \mu)(k) \to H^1_{fppf}(k,\mu)
$$  
is surjective.
\end{theorem}


The  flat cohomology (see \cite{Sh}, \cite[app. B]{BFT} or \cite{Gir}) is the right set up where to phrase the problem if the order of $\mu$
is not invertible in $k.$ If the order of $\mu$ i invertible in $k$ then the flat and usual Galois cohomology coincide.
By continuing the exact sequence of pointed sets 
$$
1 \to \mu(k) \to G(k) \to (G/ \mu)(k) \to H^1_{fppf}(k,\mu) \to H^1_{fppf}(k,G),
$$
we see that $H^1_{fppf}(k,\mu) \to H^1_{fppf}(k,G)$ is the trivial map.
In other words, the center of $G$  does not contribute to $H^1(k,G)$.\footnote{The reason why we can avoid  groups of type $E_8$ is because such groups have trivial center.}  



\subsection{Other  exceptional  groups}

\begin{theorem}\label{other} \cite{C03} \cite{Gi01} Let $G/k$ be a semisimple group satisfying
 the hypothesis of  Conjecture \ref{strong}.  
Then $H^1(k,G)=1$ in the following cases: 

\begin{enumerate}
\item  $G$ is trialitarian and its Allen algebra is of 
index $\leq 2$. 

\item $G$ is of quasi-split type $^1E_6$ or $^2E_6$ and its Tits   algebra is of 
index $\leq 3$. 

\item $G$ is of type $E_7$ and its Tits   algebra is of 
index $\leq 4$. 
\end{enumerate}

\end{theorem}

\smallskip

\noindent Furthermore the groups under consideration  are either quasi-split or isotropic  of Tits indices
$$
\begin{picture}(40,15)
\put(00,00){\line(1,0){30}}
\put(30,0){\oval(60,20)[l]}
\put(0,0){\circle*{3}}
\put(30,0){\circle*{3}}
\put(30,10){\circle*{3}}
\put(30,-10){\circle*{3}}
\put(00,0){\circle{10}}
\end{picture},
$$

$$
%\mbox{$^1E_{6,2}^{1,6}, \, ^2E_{6,2}^{1,6''} \quad$}
\quad 
\begin{picture}(80,30)
\put(00,02){\line(1,0){25}}
$$\put(65,02){\oval(80,10)[l]}
%\put(0,02){\circle*{3}}
%\put(25,02){\circle*{3}}
\put(0,02){\circle{10}}
\put(25,02){\circle{10}}
\put(65,-3){\circle*{3}}
\put(65,7){\circle*{3}}
\put(45,7){\circle*{3}}
\put(45,-3){\circle*{3}}
%\put(65,02){\oval(10,20)}
\put(0,02){\circle*{3}}
\put(25,02){\circle*{3}}
\put(-5,11){$\alpha_2$}
\put(20,11){$\alpha_4$}
\put(40,11){$\alpha_3$}
\put(62,17){$\alpha_1$}
\put(62,-15){$\alpha_6$}
\put(40,-13){$\alpha_5$}
\end{picture}
$$

$$
\begin{array}{ll}
\mbox{\rm a)}
\qquad 
\begin{picture}(105,45)
\put(00,02){\line(1,0){100}}
\put(60,02){\line(0,1){20}}
\put(00,02){\circle*{3}}
\put(20,02){\circle*{3}}
\put(40,02){\circle*{3}}
\put(60,02){\circle*{3}}
\put(80,02){\circle*{3}}
\put(100,02){\circle*{3}}
\put(60,22){\circle*{3}}
\put(20,02){\circle{10}}
\put(60,02){\circle{10}}
\put(80,02){\circle{10}}
\put(100,02){\circle{10}}
\put(-5,-11){$\alpha_7$}
\put(15,-11){$\alpha_6$}
\put(35,-11){$\alpha_5$}
\put(55,-11){$\alpha_4$}
\put(75,-11){$\alpha_3$}
\put(95,-11){$\alpha_1$}
\put(55,30){$\alpha_2$}
\end{picture}
&
\mbox{\qquad \rm  b)}
\qquad 
\begin{picture}(125,45)
\put(00,02){\line(1,0){100}}
\put(60,02){\line(0,1){20}}
\put(00,02){\circle*{3}}
\put(20,02){\circle*{3}}
\put(40,02){\circle*{3}}
\put(60,02){\circle*{3}}
\put(80,02){\circle*{3}}
\put(100,02){\circle*{3}}
\put(60,22){\circle*{3}}
%\put(00,02){\circle{10}}
%\put(20,02){\circle{10}}
\put(100,02){\circle{10}}
\put(-5,-11){$\alpha_7$}
\put(15,-11){$\alpha_6$}
\put(35,-11){$\alpha_5$}
\put(55,-11){$\alpha_4$}
\put(75,-11){$\alpha_3$}
\put(95,-11){$\alpha_1$}
\put(55,30){$\alpha_2$}
\end{picture}
\end{array}
$$

\vskip6mm

\noindent where case $a)$ (resp. $b)$) holds when the Tits algebra is of index $2$ (resp. $4$).
One more reason why other exceptional groups are difficult to deal with
is because they are anisotropic.

%\begin{corollary} \label{cor-other} \cite{CTGP}
%Let $G/k$ as in  Theorem \ref{other}. For every 
%separable finite field extension $L/k$,  assume that  every
%central simple $L$-algebras of period $2$  (resp. $3$)  is 
 % of index $\leq 2$ (resp. $3)$.
%If $G$ is trialitarian or of type $E_7$ (resp. $E_6$), then 
%$H^1(k,G)=1$.
%\end{corollary}

For function fields of surfaces, central simple algebras of period $2$ (resp.$3$) are
of index $2$ (resp. $3$)
 as pointed by Artin \cite{Ar}. Thus 
Theorem \ref{other} covers all groups of type $^{3,6}D_4$, $^{1,2}E_6$, and $E_7$ for these fields. 

In joint work with  with Colliot-Th\'el\`ene and Parimala \cite{CTGP},  we exploited Serre's conjecture II
to study  arithmetic properties of not necessarily simply connected groups. Our methods were inspired by Sansuc's paper \cite{Sa}
in the number field case. For more on this topic, see  the paper by  Borovoi and Kunyavski\v {\i} \cite{BKG}.  
 



\section{Known cases in terms of fields}

\subsection{$l$-special fields}

\noindent (a) If $l=2,3,5$ and $k$ is an $l$-special field of separable dimension $\leq 2$, Conjecture \ref{strong} holds for 
the split group of type $E_8$, see  \cite{C94} for $l=5$ and  \cite[\S III.2]{Gi01}. 

\medskip


\noindent (b) If $l=3$ and $k$ is an $l$-special field of characteristic $\neq 2$ and 
separable dimension $\leq 2$, then 
Conjecture \ref{strong} holds for trialitarian groups.
For $l=3$,  this follows of Theorem \ref{rost}.


\medskip

In both cases, a positive answer to Serre's injectivity question would provide Conjecture \ref{strong} for the groups under consideration.


\subsection{Complete valued fields}

Let $K$ be a henselian  valued field for a discrete valuation with perfect residue field
$\kappa$. A consequence of the Bruhat-Tits decomposition for Galois cohomology over complete fields
is the following. 


\begin{theorem} (Bruhat-Tits \cite[cor. 3.15]{BT3})
Assume that $\kappa$ is of cohomological dimension $\leq 1$.
Let $G/K$ be a simply connected semisimple group. 
Then $H^1(K,G)=1$.
\end{theorem}

Note that the hypotheses imply that $K$ is of separable dimension $\leq 2$.
Serre asked whether it can be generalized 
when assuming $[\kappa:\kappa^p] \leq p$ \cite[5.1]{Se2}.
The hypothesis $[\kappa:\kappa^p] \leq p$ alone is 
not enough  here because $K=\bF_p((x))((y))$ is of separable dimension $3$ 
and is complete with residue field
$\bF_p((x))$. 



But if $\kappa$ is separably closed and $[\kappa:\kappa^p] \leq p$, then $K$  
is of separable dimension $1$   and enough cases of the vanishing of $H^1(\kappa((x)), G)$ have been established
in view of the proof of Tits conjectures on unipotent subgroups \cite{Gi02}.
The general case, however,  is still open.







%This is actually a special case of Conjecture \ref{strong}
%since $K$ is then a field of separable dimension $2$.
%For $l \not =p$, this is \cite[]{GSM}.
%Indeed, if $K$ is of characteristic zero,
%Kato's filtration of  $H^3_p(K)$ shows that $H^3_p(K)=0$

%Indeed, if $K$ is of positive characteristic, $[K;K^p] \leq p^2$ and then $H^3_p(K')$.


Note also that the conjecture is proven for fraction fields of
 henselian two dimensional local rings  with algebraically closed
 residue field of characteristic zero, e.g. $\C[[x,y]]$, as shown in  \cite{CTGP}. For the  $E_8$ case, a key point is that 
the derived group of the absolute Galois group
is of cohomological dimension $1$ \cite[Th. 2.2]{CTOP}. 



\subsection{Global fields} The number field case is due to Kneser for classical groups \cite{Kn}, Harder for
exceptional groups of type other than  $E_8$ \cite[I, II]{Harder}, and Chernousov for type $E_8$ \cite{C89}, see also \cite{PR}.
The function field case due to Harder \cite[III]{Harder}. 



\subsection{Function fields}

He, de Jong and Starr have proven   Conjecture \ref{strong} for split groups over  function fields
in a uniform way in arbitrary characteristic.

\begin{theorem}  \label{main}  \cite[cor. 1.5]{dJHS}
Let $k$ be an algebraically closed field and let  $K$ be the function field of 
a quasi-projective smooth surface $S$. Let $G$ be a split semisimple simply connected group over $k$. Then $H^1(K,G)=1$.
\end{theorem}

Except $E_8$, all other cases of the conjecture were known by  case by case 
considerations \cite{CTGP}.
Hence Conjecture \ref{strong} is fully proven for function fields of surfaces.
The proof of Theorem \ref{main} is based on 
the existence of sections for fibrations in rationally simply connected varieties.\footnote{ Integral versions of this result have recently been studied. The case when the function field $\Bbb C (t_1, t_2)$ is replaced by the Laurent polynomial ring  $\Bbb C [t_1^{\pm 1}, t_2^{\pm 1}]$ appear in the study of infinite dimensional Lie algebras. In analogy with conjecture II it is natural to ask whether $H^1_{fppf}(\Bbb C [t_1^{\pm 1}, t_2^{\pm 1}],  \mathfrak{G})$ vanishes for all simple simply connected group schemes $\mathfrak{G}$ over $\Bbb C [t_1^{\pm 1}, t_2^{\pm 1}].$ An example of B. Margaux shows that the answer to this question is negative (see \cite{GP07} \S 3.6.) }


\begin{theorem} \label{step-main}  \cite[Th. 1.4]{dJHS}
Let $S/k$ as in Theorem \ref{main}. Let $X/S$ be a projective morphism whose
geometric generic fiber is a twisted flag variety. Assume that
$\Pic(X ) \to \Pic( X \times_K \overline K)$ is surjective.
Then $X \to S$ has a rational section. 
\end{theorem}

The assumption on the Picard group means that there is no ``Brauer obstruction''.
When applied to higher Severi-Brauer schemes, this statement yields as a corollary
de Jong's theorem ``period=index'' \cite{dJ} for central simple algebras over such fields; see also
\cite{CTdJ}. This is the first classification-free work described in this survey. 


\subsection{Why Theorem \ref{step-main} implies Theorem \ref{main}}
We take  this  opportunity to reproduce how this argument goes.



\begin{lemma} \label{panin} Set $G/F$ be a semisimple simply connected group 
over a field $F$. Let $E/F$ be a $G$-torsor.

\begin{enumerate}
\item $\Pic(E)=0$ and we have an exact sequence 
$$ 
0 \to \br(F) \to \br(E) \to \br(E \times_F F_s). 
$$
\item Let  $P$ be a $F$-parabolic subgroup of  $G$ and let 
$E/P$ the variety of  parabolic subgroups of the twisted $F$-group $E(G)$
having the same type than $P$.
Then $\Pic(P)=0$ and we have an exact sequence  $$
0 \to \br(F) \to \br(E/P) \to \br(E/P \times_F F_s)
$$   and 
an isomorphism
 $\Pic(E/P) \simlgr  \Pic(E/P \times_F F_s)^{\Gal(F_s/F)}$.
\end{enumerate}

\end{lemma}

\begin{proof} (1)
We have $H^1(F, (F_s)^\times)=0$ and $\Pic(E \times_F F_s)
\cong  \Pic(G \times_F F_s)=0$ since $G$ is simply connected \cite{FI}. The first  terms
of the Hochschild-Serre spectral sequence
$H^p(\Gal(F_s/F), H^q( E \times_F F_s, {\bf G}_m)) \Longrightarrow
H^{p+q}( E, {\bf G}_m)$ show that 
$\Pic(E)=0$ and that the sequence
 $0 \to \br(F) \to \br(E) \to \br(E \times_F F_s)$ is exact.

\smallskip

\noindent {$2)$} The fibration $G \to G/P$ is locally trivial for the Zariski topology (Borel-Tits). 
By a result of Sansuc applied to the fibration $G \to G/P$ 
 \cite[6.10.2]{Sa},\footnote{Strictly speaking, Sansuc's result is proved
under the assumption that $k$ be perfect or that the group of the fibration be reductive.
By d\'evissage, one can see that the result holds for fibrations under  groups that are extensions of 
reductive $k$-group by split  $k$--unipotent groups.}  we have a surjective map $\Pic(G) \to \Pic(P)$, hence 
$\Pic(P)=0$.
 By \cite[6.10.1]{Sa} applied to the fibration $E \to E/P$, there is
an exact sequence
$$
0  =  \Pic(P) \to \Br(E/P) \to \Br(E),
$$
hence the map  $\br(E/P) \to \br(E)$  is injective. We look  at the commutative diagram
$$
\begin{CD}
&& &&0  \\
&&&& @VVV \\
 0 @>>> \br(F) @>>>   \br(E/P) @>>> \br(E/P \times_F F_s) \\
&&\mid \mid && @VVV @VVV \\ 
 0 @>>>  \br(F) @>>>   \br(E) @>>> \br(E \times_F F_s) .\\
\end{CD}
$$
Since the bottom sequence is exact, we get by diagram chasing that the upper horizontal
sequence is exact as well. The second isomorphism $\Pic(E/P) \simlgr  {\Pic(E/P \times_F F_s)^{\Gal(F_s/F)}}$ comes from the 
Hochschild-Serre spectral sequence.
\end{proof}

For complete results on Picard and Brauer groups of twisted flag varieties, see Merkurjev-Tignol \cite[\S 2]{MT}.


\begin{proposition} \label{step-main2}  \cite[Th. 1.4]{dJHS}
Let $S/k$ as in Theorem \ref{main}. Let $G/K$ be a 
 simple simply connected $K$-group which is an inner form and let $P$ be 
 a $K$-parabolic subgroup of $G$.
Then the map $H^1(K,P) \to H^1(K,G)$ 
is bijective.
\end{proposition}


Proposition \ref{step-main2} implies Theorem \ref{main} by taking a Borel
subgroup of $G$ because $H^1(K,B)=1$.

\begin{proof} Injectivity is a general fact due to Borel-Tits (\cite{BoT}, th\'eor\`eme 4.13.a).
Let  $E/K$ be a $G$-torsor of class $[E] \in H^1(K,G)$.
Up to shrinking $S$, we can assume that $G/K$ extends to 
a semisimple group scheme $\GB/S$, $P/K$ extends to a $S$--parabolic subgroup scheme
$\PB/S,$ and that  $E/K$ extends to a $\GB$-torsor $\E/S$ \cite{Ma}. 
By \'etale descent, we can twist the $S$-group scheme  $\GB/S$
by inner automorphisms, namely we can define the $S$-group scheme
 $\E(\GB)/S$.
We define then  $\V/S:= \E/\PB$, i.e. the scheme of 
parabolic subgroup schemes of $\E( \GB) /S$ (\cite{SGA3}, exp. XXVI) of the same type 
than $P$.
The  morphism $\pi :\V \to X$ is projective, smooth  and with geometrically
integral fibers.
Set $V= \V \times_S K$, this is a generalised twisted flag variety.
Since  $G$ is assumed to be an inner form,
$\Pic(V \times_K K_s)$ is a trivial $\Gal(K_s/K)$-module.
By Lemma \ref{panin}.2, the map
$$
\Pic(V ) \to \Pic(V \times_K K_s) 
$$
is onto. Thus the composite map 
$\Pic(\V)  \to \Pic(V ) \to \Pic(V \times_K K_s)$ is onto. 
Theorem \ref{main} applies and shows that $V(K) \not= \emptyset$.
It means that the  torsor $E$ admits  a reduction to $P$
 (\cite{Se1}, \S I.5, proposition 37), that is 
$[E]\in {\rm Im}\bigl(  H^1(K,P) \to H^1(K,G) \bigr)$.
We conclude that the mapping   $H^1(K,P) \to H^1(K,G)$ is surjective.
\end{proof}



The  Grothendieck-Serre's conjecture on rationally trivial torsors has been proven
by Colliot-Th\'el\`ene and Ojanguren 
for  torsors over a semisimple group  defined over an algebraic closed field \cite{CTO}.
Thus He-de Jong-Starr's theorem has the following geometric application.


\begin{corollary}\label{cor-main} Let $S/k$ be a smooth quasi-projective surface.
Let $G/k$ be a (split)  semisimple simply connected group. Let $E/S$ be a $G$-torsor.
Then $E$ is locally trivial for the Zariski topology.
\end{corollary}




\section{Remaining cases and open questions}


$\bullet$ Provide  a classification free proof of 
the case of totally imaginary  number fields, at least in the quasi-split case.

\medskip

$\bullet$ The first remaining cases of Conjecture \ref{strong}  are that of trialitarian groups, 
groups of type $E_6$ over a $3$--special field, groups of type
$E_7$ over a $2$-special field and groups of type $E_8$.

\medskip



$\bullet$ Let $K$ be a function field  of surface over an algebraically closed field.
Are $K$--division algebras cyclic ?
Is it true that $\cd(K_{ab}) =1$ where $K_{ab}$ stands for the abelian closure of $K$ ?

\smallskip

In the global field case, class field theory  answers  both   questions positively.
The question on $K_{ab}$ is due to Bogomolov and  makes sense for arbitrary fields.
As noticed by Chernousov, Reichstein and the author, a positive answer would
provide a positive answer to Serre's conjecture II for groups of type $E_8$  \cite{CGR}. 

\medskip


$\bullet$  For the Kneser-Tits conjecture for perfect fields of cohomological dimension $\leq 2$,
there remains only the case of a group of Tits index \cite[\S 8.2]{Gi07} 
$$
\mbox{$E_{7,1}^{66} \qquad$}
\begin{picture}(105,45)
\put(00,02){\line(1,0){100}}
\put(60,02){\line(0,1){20}}
\put(00,02){\circle*{3}}
\put(20,02){\circle*{3}}
\put(40,02){\circle*{3}}
\put(60,02){\circle*{3}}
\put(80,02){\circle*{3}}
\put(100,02){\circle*{3}}
\put(60,22){\circle*{3}}
\put(20,02){\circle*{3}}
\put(60,02){\circle*{3}}
\put(80,02){\circle*{3}}
\put(100,02){\circle{10}}
\put(-5,-11){$\alpha_7$}
\put(15,-11){$\alpha_6$}
\put(35,-11){$\alpha_5$}
\put(55,-11){$\alpha_4$}
\put(75,-11){$\alpha_3$}
\put(95,-11){$\alpha_1$}
\put(55,30){$\alpha_2$}
\end{picture}
$$


\bigskip

\bigskip






\bigskip


\noindent{\it Adress : DMA, UMR 8553 du CNRS, Ecole normale sup\'erieure, 45 rue d'Ulm}

\noindent{\it F-75005 Paris, France.} 

\end{document}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\section{Fields of virtual cohomological dimension 2}

Colliot-Th\'el\`ene made the following conjecture which generalizes 
the Hasse principle \cite[Conjecture 2.11]{CT96}.


\begin{conjecture}\label{conjIII} Let $k$ be a field of characteristic zero 
such that ${\rm cd}_l(k (\sqrt{-1}) \leq 2$ for all torsion primes of $G$.
Let $G/k$ be a semisimple simply connected group. Then the map
$$
H^1(k,G) \to \prod\limits_{v} H^1(k_v,G)
$$
 is injective, where $v$ runs over the ordering of $k$ 
and $k_v$ stands for the real closure pf $k$ with respect to an ordering $v$.
\end{conjecture}

The condition ${\rm cd}_l(k (\sqrt{-1}) \leq 2$ is equivalent to require  that
the absolute Galois group of $k_l$ admits an open subgroup which is 
of cohomological dimension $\leq 2$.  This is the reason why we speak
about a field of virtual cohomological dimension $\leq 2$.


The  positive characteristic case is irrelevant here 
since the $l$-cohomological dimension does not change in that case by finite extension
 and there are no orderings by Artin-Schreier \cite[II.4.1]{Se1}.

Beyond the Hasse principle (Kneser \cite{Kn}, Harder \cite{Harder}, Chernousov), the main evidence is the following result
by Bayer-Parimala.


\begin{theorem} \cite{BP2} Under the hypothesis 
of conjecture \ref{conjIII}, assume that  $G$ is  classical 
or of type $G_2$ or $F_4$. 
Then Conjecture \ref{conjIII} holds for $G$.
\end{theorem}


The inner type $A$ is crucial. Precisely, given a central simple algebra $A$, 
 a scalar $c\in k^\times$ is a reduced norm if and only if
$c$ belongs to $\nrd(A \otimes_k k_v)$ for all
orderings $v$ or equivalently if $c$ is postive with respect to $v$
for all orderings $v$ such that $A \otimes_k k_s$ does not split.
This result is linked with the classification. A semisimple group $G/K$
is classified by his quasi-split form, its Tits algebras and the $G \times_K K_v$.
It is applied in hermitian forms and for study on of the groups \cite{KP}. 

For exceptional groups, we have the following result.

\begin{theorem} (Chernousov, \cite[th. 7.2]{C03})
 Let $G$ be a quasi-split semisimple group of Cartan-Killing type
$D_4$, $E_6$ or $E_7$ defined over a perfect field $k$ of
virtual cohomological dimension $2$.  Then Conjecture \ref{conjIII} holds for $G$.  
\end{theorem}



%%\subsection{Link with Bogomolov's conjecture}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



\begin{proof} Let $O$ be the valuation ring of $K$ and denote by
$O_{nr}/O$ the  extension of henselian local rings  
 associated to $\kappa_s/\kappa$.
The fraction field $K_{nr}$ of $O_ {nr}$ is called  the unramified closure of $K$
and is of Kato dimension $\leq 1$ (REF).
By the extension of Steinberg's theorem \cite[th. 11.12]{St} by 
Borel-Springer \cite[\S 8.6]{BoS}, we know that
$H^1(K_{nr},G)=1$ (see also \cite[\S 4.2, th. 4']{Se2}. Hence 
$$
H^1\bigr(\Gal(\kappa_s/\kappa), G(K_{nr}) \bigl) \simlgr H^1(K,G).
$$ 
Apply Bruhat-Tits decomposition {\bf COMPLETE}.
\end{proof}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



